Author’s response to reviews

Title: Association between egg consumption and elevated fasting glucose prevalence in relation to dietary patterns in selected group of Polish adults

Authors:

Anna Czekajło-Kozłowska (anna.czekajlo-kozlow@umed.wroc.pl)

Dorota Rozanska (dorota.rozanska@umed.wroc.pl)

Katarzyna Zatońska (katarzyna.zatonska@umed.wroc.pl)

Andrzej Szuba (szubaa@yahoo.com)

Bożena Regulska-Ilow (bozena.regulska-ilow@umed.wroc.pl)

Version: 3 Date: 17 Dec 2019

Author’s response to reviews:

Dear Editors of Nutrition Journal,

We thank the Reviewer for the constructive comment on our manuscript. The manuscript has been revised and all changes are highlighted in yellow.

Answers to Reviewer #2:

1. I recommend improving conclusions (both in the abstract and main text) to give more clear information to readers.

Response: The conclusions section was carefully revised and corrected both in the abstract and main text.

I hope that the revised manuscript will be accepted for publication in Nutrition Journal.

Yours sincerely,

Dorota Różańska